Author's response to reviews

Title: Prevalence and determinants of osteoporosis in patients with type 1 and type 2 diabetes mellitus: a cohort study

Authors:

- Gudrun Leidig-Bruckner (thomas.bruckner@t-online.de)
- Sonja Grobholz (sgrobholz@gmx.de)
- Thomas Bruckner (bruckner@imbi.uni-heidelberg.de)
- Christa Scheidt-Nave (scheidt-navec@rki.de)
- Peter Nawroth (pater.nawroth@med.uni-heidelberg.de)
- Jochen G Schneider (jochen.schneider@mailbox-js.de)

Version: 2
Date: 18 November 2013

Author's response to reviews:

Dear Ms. Rada, dear Editors,

please find enclosed the revised version of our manuscript, titled "Prevalence and determinants of osteoporosis in patients with type 1 and type 2 diabetes mellitus: a cohort study".

We have performed the required changes:

The email addresses of all authors were added in the title page.

A section "Acknowledgements" was added.

We apologize for the formal errors and hope that the paper is now eligible for peer review process.

With kind regards

Gudrun Leidig-Bruckner